Author's response to reviews

Title: Retear of Anterior Cruciate Ligament Graft in Female Basketball Players

Authors:
Yoshinari Tanaka (yoshi-t@fa3.so-net.ne.jp)
Yasukazu Yonetani (yonetani@orh.go.jp)
Yoshiki Shiozaki (yks@ares.eonet.ne.jp)
Takuya Kitaguchi (reha-pt@orh.go.jp)
Nozomi Sato (reha-pt@orh.go.jp)
Shinya Takeshita (reha-pt@orh.go.jp)
Shuji Horibe (horibe@orh.go.jp)

Version: 3 Date: 4 March 2010

Author's response to reviews: see over
Dear SMARTT Editorial Team:

Thank you very much for your E-mail of March 3, 2010, concerning our manuscript (MS#2110354418321805) entitled “Retear of Anterior Cruciate Ligament Graft in Female Basketball Players” by Y. Tanaka et al. We carefully ensured and made following final changes according to your suggestion.

Please let us know if there are any problems.

Sincerely yours,

Yoshinari Tanaka, M.D., Ph.D.
Department of Sports Orthopaedics,
Osaka Rosai Hospital
Osaka, JAPAN
Major revisions

Title, Abstract
We ensured that the title and abstract are identical in the manuscript and the submission system.

Authors’ contribution
The all authors approved the final manuscript.

Minor revisions

Title:
We changed the title to capitalize only the first word.

Key words:
We removed.

E-mail address:
E-mail addresses of all authors were added in the title page.

References:
We changed them according to your offer.

Figures:
According to your comment, we changed the images of Fig 1 and 2 for new ones to remove white space.

Abstract (page 2, lines 23-24)
We changed the final sentence in the abstract “Closer attention should be paid to player preoperative condition, including muscle strength and postoperative status” to “Closer attention should be paid to player preoperative condition, as well as muscle strength and postoperative status”.